Author’s response to reviews

Title: Professionals' experiences with pediatric colonoscopy: an interview study (Original Article)

Authors:

Vedrana Vejzovic (vedrana.vejzovic@mau.se)

Version: 2 Date: 30 Nov 2018

Author’s response to reviews:

BMC Nursing
Editor Tovah Honor Aronin

Manuscript NURS-D-18-00084R2 entitled "Professionals' experiences with paediatric colonoscopy: an interview study (Original Article)"

Thank you for the reviewers' valuable comments. The modified text, after the reviewers comments, is highlighted in red in the main document and my response to the reviewers can be found in this document under "Response".

Reviewer comments: While the content of the manuscript has certainly been improved and clarified, there are still some elements requiring attention. In the abstract, the introduction is unnecessarily long, while the results section is rather short and could include a more thorough precis of the findings. The introduction to the main paper is also quite lengthy and detailed. Some extraneous information could be removed, or perhaps shifted into the discussion. The English language would still benefit from some further review and revision. It still comes across as rather 'clunky' throughout.

Response: In the abstract, the Introduction has been abbreviated, the introduction in the main paper as well. The English language has been revised by American Journal Experts AJE.